Author’s response to reviews

Title: A systematic review of the diagnostic accuracy of prostate specific antigen

Authors:

Philip R Harvey (Pxh553@bham.ac.uk)
Amman Basuita (Axb526@bham.ac.uk)
Deborah R Endersby (Dxe566@bham.ac.uk)
Ben Curtis (Bxc544@bham.ac.uk)
Aphrodite Iacoviadou (Axi520@bham.ac.uk)
Mary E Walker (Mew506@bham.ac.uk)

Version: 6 Date: 31 August 2009

Author’s response to reviews: see over
Dear Editor,

Thank you for accepting the manuscript. The authors have made all the major amendments.
1. Title, authors list, affiliations and authors contribution have been added and updated.
2. The additional files have been reformatted and named as requested.
3. The reference list is now part of the main document, the numbering system has been corrected from superscript to square bracketed numbers and reference five has been corrected.
4. Figure titles have now been moved to the end of the document.
5. Superscript has been used to link the authors with there affiliations.
6. The abstract has been appropriately titled “abstract”
7. The author’s conclusions section has been renamed conclusions.

Please get in touch if there are any further major formatting changes required.

Many thanks,

Phil Harvey